

Dear Editors:

Thank you very much for your review of our manuscript entitled “ **Emergent Single Balloon Enteroscopy for Overt Bleeding of Small Intestinal Vascular Malformation**” (Manuscript NO: 36029). We have made substantial amendments of the manuscript according to the valuable suggestions of the reviewers. Our detailed point-by-point responses are given below. We also highlighted the changes with red font in the revised manuscript. We hope our revision is satisfactory and the style of the manuscript has been met. Thank you for your consideration of a revised version of our manuscript.

Yours Sincerely,

Chen-Shuan, Chung, MD, MSc

Division of Gastroenterology and Hepatology, Department of Internal Medicine, Far Eastern Memorial Hospital, Banciao District, New Taipei City, Taiwan

No. 21, Section 2, Nanya South Road, Banciao District, New Taipei City, Taiwan

Phone: +886-8966-7000 ext 1704

Fax: +886-8966-5567

E-mail: chungchenshuan\_3@yahoo.com.tw

## **Reviewer**

According to the last ASGE guidelines, enteroscopy allows not only for diagnosis and biopsy but also for therapeutic interventions. In this case, the single balloon enteroscopy was efficient to allow the diagnosis and an economic enterectomy assisting a laparoscopic segmental resection of the jejunum, once the lesion's localization was made by the angiography. Indeed the angiography is the best approach when we have an unstable scenario. In the introduction the authors described the therapeutic approach. I suggest re-writing this paragraph, showing the advantages of diagnosis since this case refers to diagnosis, not the therapeutic procedure which was done by surgery. I would also like that the authors discuss more about the single-balloon enteroscopy after radioscopy procedure.

**Re:** Thank you for your critical review. We had made amendment about the "Introduction" part and emphasized on the role of deep enteroscopy in the diagnosis of small intestinal diseases. We also discuss more about the approach method by enteroscopy after radiological intervention in the "Discussion" part.